

Dear Editors and Reviewers:

Thank you for your letter and for the reviewers' comments concerning our manuscript entitled "Remission of hepatotoxicity in chronic pulmonary aspergillosis patients after lowering trough concentration of voriconazole".

Based on your comment and request, we have made modification on the original manuscript. A document answering every question from the referees was also summarized and enclosed.

We appreciate for Editors/Reviewers' warm work earnestly, and hope that the correction will meet with approval.

Once again, thank you very much for your comments and suggestions

Sincerely yours,

Xiuhong Nie

Response:

Our deepest gratitude goes to the reviewers for their careful work and thoughtful suggestions that have helped improve this paper.

Answer:

(1) Science Editor : (1) The authors did not provide original pictures. Please provide the original figure documents. Please prepare and arrange the figures using **PowerPoint** to ensure that all graphs or arrows or text portions can be reprocessed by the editor; (2) PMID numbers are missing in the reference list. Please provide the PubMed numbers to the reference list and list all authors of the references. Please revise throughout; and (3) The “Article Highlights” section is missing. Please add the “Article Highlights” section at the end of the main text. 6 Re-Review: Not required. 7 Recommendation: Conditional acceptance.

- For drawings that were automatically generated with software (SPSS software), we provide the labels/values of the ordinate.
- We have supplemented the PubMed numbers to the reference list and list all authors of the references.
- We have written the “article highlight” section.

(2) Editorial Office Director : The author should fill out each item of the STROBE checklist with page numbers.

- We filled out each item of the STROBE checklist with page numbers.

(3) Company Editor-in-Chief : the full text of the manuscript, the relevant ethics documents, and the English Language Certificate, all of which have met the basic publishing requirements of the World Journal of Clinical Cases, and the manuscript is conditionally accepted.

- Thank you very much for your appreciation.